Author’s response to reviews

Title: Development and evaluation of a real-time one step Reverse-Transcriptase PCR for quantitation of Chandipura Virus

Authors:

Satyendra Kumar (prashersk@gmail.com)
Ramesh S Jadi (ramesh_pune@yahoo.com)
Sudeep B Anakkathil (sudeepmcc@yahoo.co.in)
Babasaheb V Tandale (drtandale@gmail.com)
Akhilesh Chandra Mishra (acm1750@rediffmail.com)
Vidya A Arankalle (varankalle@yahoo.com)

Version: 5 Date: 6 December 2008

Author’s response to reviews: see over
Dear Dr. Rikki Graham,
Senior Assistant Editor
BMC-series journals
Tel: +44 (0)20 7631 9921
Facsimile: +44 (0)20 7631 9923
e-mail: editorial@biomedcentral.com
Web: http://www.biomedcentral.com/

Thank you very much for giving us the final opportunity to review our manuscript entitled “Development and evaluation of a real-time one step Reverse-Transcriptase PCR for quantitation of Chandipura Virus” by Kumar et al. We have done some changes/modifications as suggested by your production team. We are also providing point wise clarification for these changes.

**Point of clarification:**
1. Abstract: Page 2, Results, line-6 ‘suckling mice’ should read “infant mice”.
2. Background: Page 3 line 28, ‘real-time PCR’ should read “real-time one step RT-PCR”.
4. Methods: Page 5, line 29, ‘GM’ should read “Grace’s insect cell culture media”.
5. Methods: Page 6, line 18, ‘suckling mice’ should read “infant mice”.
6. Discussion: Page 9, line 10, ‘suckling mice’ should read “infant mice”.

I hereby confirm that all the authors have participated in the revision and subsequent submission. We do hope that the revised version will be suitable for the publication in an early issue of the BMC Infectious Diseases journal.
Thanking you

Yours Truly

Vidya A Arankalle
Scientist ‘F’ and Head
Hepatitis Division
National Institute of Virology
130/1, Sus Road, Pashan
Pune, India, 411021
Phone: +91-020-2612-7301
Fax: +91-020-2612-2669
Email: varankalle@yahoo.com
nivhep@gmail.com